
 
 
 
 
 
 
 
 
 
 
 
 
 
 
 

 
ARM Nauru Research Station 

Site Visit 0509N Report 
 

Visit Duration: 15 September to 27 September 2005 
 

Nauru Island, Republic of Nauru 
 
 
 



Priority

System 
Date 

Completed 
(dd/Month)

Service Time 
in Hours

Tasks/Assoc Hazards-Controls 
For each priority item insert new line(row) for each date/time entry

1 AERI Install repaired system
16-Sep 4 Clean out Aeri Van to remove dust and mould
17-Sep 12 Finished Van cleanout - walls scrubbed with bleach.  Fluro light 

replaced - old one faulty.  Commenced installation of the Aeri.  
Aeri electronics rack assembled in shelter, interferometer 
assembled and connected to Aeri electronics rack.   Basic testing 
commenced to confirm system is functioning.

18-Sep 6 Continued with Aeri preliminary tests.  Sun sensor does not 
appear to be operating correctly and the internal rain sensor is 
suspect (still to confirm the rain sensor).  Email sent to Ralph to 
query what to further check.  Reseated J20/J21 connectors and 
tested the rain and sun detectors again and all appeared to now 
be functioning - advised Ralph.

19-Sep 2 Review proceedure for FOV checks on Aeri
23-Sep 9 FOV mapping and 3 black body comparisons done on Aeri.  Data 

sent to Denny and Ralph @ WISC
24-Sep 8 WISC requested 3BB tests be done again.  
25-Sep 4 Packup of test equipment, and general tidy up of van
26-Sep 8 3BB tests being done again for WISC

2 AERI Change out blower belt
18-Sep 1 Blower belt changed and bearing alignment checked

3 MWR Troubleshoot and repair
25-Sep Waiting replacement MWR 

Site Tasks

SV-0509N
Proposed Dates on Site: 15 - 27 September 2005
Ship or hand carry prior to SV:
AERI; ventilation fans for Dell PCs; 1 bottle of Envirogel (hand carry); Capacitors 
for NEC split A/C units (hand carry); Batteries for AERI and electronics - backed 
up by APC SU2200NET, fed by non-UPS 120V AC (handcarry or ship); Batteries 
for BBSS - backed up by APC BP420S, fed by non-UPS 120V AC (handcarry or 
ship)

Site:

Safety:
Major Hazards: Caustic chemicals (KOH), Explosive gas (H2), AC electricity

Hazard Controls: BOM Electrolyser Handbook, BM Electrolyzer 
Operations/Maintenance Training, ASNZS 4836/2001-Safe Working on LV 
Electrical Installations, 2-man rule

Team Members:
Rex Pearson
Troy Culgan
Paul Foote

Reminder:
When replacing items (e.g., desiccant, filters, etc.), record and report the quantity 
of spares left on site.



4 MFRSR Troubleshoot and repair
15-Sep 4 MFRSR motor totally seized.  Removed logger box from stand in 

an attempt to open the lid of the box (2 of the screws were 
seized).  Lid repaired and new motor installed.  MFRSR initialised 
and is running.  Will need to adjust shadowband when sunny.

16-Sep 1 Shadow band adjusted and rechecked at solar noon

5 TSI Troubleshoot
15-Sep 4 Remove camera assembly, check for water damage - ok. 

Remove ND filter and clean - salt buildup all over the lens.  Clean 
camera lens.  Refocus camera and install ND filter.

16-Sep 1 Adjusted camera arm to extreme RHS position to centre image 

18-Sep 0.5 TSI rechecked and functioning normally

6 CAL Record configurations before and after cals
18-Sep 2 Initial .CSI files uploaded to TWP FTP site for Sky,Gnd & Smet

23-Sep 3 Continued configuration records
26-Sep 1 Final records completed

7 CAL Provide Cal Kit inventory list

8 DAQ Calibrate all Instrument loggers and spares
22-Sep 9 Check loggers with RSU

9 CAL Calibration/comparison skyrad radiometers
16-Sep 8 Installed radiometers and logger on to comparison stand, setup 

logger net Laptop to monitor and collect data, install networking 
from stand into D Van.

17-Sep 6 Fault finding cal logger setup, replaced cal logger 2 due to PIR1 
reading low - replacement logger solved problem.  Commenced 
collecting data.

19-Sep 8 Continued comparison checks
20-Sep 4 Instruments swapped as per Bill Porch's email
21-Sep 3 Swapped faulty NIP for comparison - taking data
24-Sep 4 Removed radiometers & cal logger from stand, stored 

radiometers in their cases.

10 CAL Calibration/comparison grnrad radiometers
17-Sep 1 Gndrad instruments inverted at 0152z
23-Sep 2 Blackbody checks on GNDRAD PIR
24-Sep 2 Gndrad restored to normal

11 SMET Calibration/comparison smet instruments and logger
20-Sep 4 Calibration commenced.
21-Sep 7 Calibration completed

12 IRT Calibration/comparison GNDRAD IRT
18-Sep 1 Gndrad IRT comparison completed



13 IRT Calibration/comparison SKYRAD IRT
18-Sep 1 Skyrad IRT comparison completed

14 CAL Place Cal/replacement records on FTP site
21-Sep 4 Initial set of records completed and data placed on site 
24-Sep 2 Completed

15 LICOR Install Licor at the small site as per Chuck Long's request
19-Sep 4 Commenced assembly of components, power supplies etc
20-Sep 2 Completed assembly and under test before placement at site.  

Sent Chuck Long an email re calibration of the licors.
21-Sep 4 Chuck Long comfirmed comparison proceedure, checking Licors 

against unshaded PSP radiometers.  Checking power supply 
system to ensure charging capability.

22-Sep 6 Licors installed at Airport site, and collecting data.   Have to install 
shelf to hold equipment.

26-Sep 3 Completed installation, collected first data and placed it on the 
twp ftp site under private/airport licor site

16 VAN A/C maintenance
???

17 GENSET Routine maintenance
19-Sep Genset mechanic arrived on site and commenced maintenance

20-Sep Genset maintenance continued
21-Sep Genset maintenance completed

18 MPL Inveistigate double-pulsing problem
18-Sep 1 Checked MPL data as per Albert Mendoza information.  MPL is 

currently not double pulsing

19 MPL Program sunshade timer
15-Sep 1 Check timer programming. Monitored shade operation at preset 

times and it functioned correctly.  Will monitor over the next few 
days.

16-Sep 0.2 Sunshade operation checked at solar noon - ok
17-Sep 0.2 Sunshade operation checked at solar noon - ok
18-Sep 0.5 Moved position of sun sensor to more accurately reflect incoming 

sunshine at solar noon
19-Sep 0.5 Sunshade operation checked at solar noon - ok

20 MPL Change out two shutter timer batteries (AA)
24-Sep 0.5 Batteries changed, terminals cleaned and shutter timer adjusted

21 VANS Install new capacitors on NEC split A/C units
 

22 SMET Install counter weight
20-Sep 2 Smet counter weights installed



23 COMM Troubleshoot e-mail attachment problem
19-Sep 4 Unable to resolve all the problems with the observer computer.  

Saved any relevant documents and formatted hard drive and 
installed a fresh copy of Windoze and commenced service pack 
updates.

20-Sep 4 Completed installation of software, configure email etc.  Noticed 
that the email service from Hotmail is continually timing out and 
the observers are unable to get attachements.  Changed account 
and setup VPN connection to Darwin to allow email thru a pop 
server.  This functioned ok.

24 AWS Change out fan in AWS SAT Phone antenna

25 Other Inspect cables per Cable Inspection Procedure
25-Sep 2 Cables checked

26 CIMEL Change out belt
25-Sep 1 Inspected belt ok

27 ES&H Take inventory for replacing existing fire extinguishers
16-Sep 1 Fire extinguisher in Z Van is showing signs of rust damage to the 

bottom of the unit.  All are normal on levels

28 ES&H Change out H-Van eye wash fluid
25-Sep 1 Changed fluid

29 ES&H Change out U-Van eye wash fluid
25-Sep 0.5 Changed fluid

30 GPS SUOMI Change out aspirator fan
22-Sep 2 Unable to locate fan - can it be confirmed one was sent for the 

Suomi

31 BBSS Change out Velcro strap sonde retainer
25-Sep 0.5 Changed out

32 IRT Change out SKYRAD IRT Gold Mirror
17-Sep 0.5 Mirror changed out and old mirror prepared for shipment.

33 MFRSR Check head and board and assess/inspect for possible 
changeout

16-Sep 2 MFRSR subsystems checked and cleaned.  No visible signs of 
damage

34 MMCR Change out antenna heater filters

35 MMCR Change out TWTA desiccant
24-Sep 0.5 Changed

36 T/RH Check handheld probe and assess/inspect for possible changeout

24-Sep 0.5 Handheld probe does not exist.  Was unreliable several years 
ago and the observers use the AWS data



37 DS Install R1 computer
16-Sep 2 R1 computer installed.  Networking setup.  Ron Reed setup login 

accounts.  A problem exists with the x driver not recognising the 
local monitor - will look into this later.

38 OTHER Install ventilation fans on Dell PCs
18-Sep 1 Replaced fans and checked capacitors on motherboards.

39 SHIPPING List items to be shipped following visit
NIP (to Tom Stoffel of NREL via SGP)

40 CIMIL Check and reset clock
17-Sep 0.5 Check and reset clock on the cimel

41 ORG Check moisture in lens
20-Sep 2 Remove cover from transmitter, clean out moisture, insert 

desiccant packs and reseal unit.

42 GC-25  Faulty GC25 ground check set
20-Sep GC25 faulty - appears to have failed due to a faulty sonde 

dragging down the power supply.  Replaced with spare unit and 
BBSS is running again.  We received RMA 85-1606 from Vaisala 
for the return authorisation for repair.

43 OH&S Install tap on water tank near E Van
22-Sep Tap installed on tank

44 AWS Cal and check
23-Sep 4 AWS cal checked - raingauge microswitch faulty, replaced and 

checked ok

45 Electrolyser Electrolyser maintenance
25-Sep 4 Electrolyser maintenance, changed head, replaced compressor 

oil etc

46 Site 
maintenance

25-Sep 2 Setup of water pressure cleaner and showed observers use, they 
are going to clean mould off all vans, treat rusting screws with 
rust inhibitor and cover with paint.

47 Manus Email Configure Manus observer computer to use new email account

25-Sep 2 Downloaded OpenVPN software, client config and key to the 
Manus observer computer.  Configured Tap and port in config file.
Configured Outlook to use both the Daltron and Darwin mail 
servers to receive mail and to use the Darwin server as the 
default send account.

48 K&Z tracker Check setup
26-Sep 1 Checked time and level adjustments



49 FUTURE 
NEEDS

List items needed for future visits (spares, etc.)

16-Sep Replacement multimeter (Fluke series) apparently this has been 
missing for months

16-Sep Fire extinguisher required for Z Van
18-Sep 1 Replacement battery pack for APC UPS in data system (Smart 

UPS 3000).  Alarm indication flashing on unit.  As this is running 
behind the Liebert UPS this should be ok except it will possible 
not provide a clean shutdown command to the DS should all 
power totally fail.  Would recommend battery packs in Manus also 
be changed at the next reset.

18-Sep 1 Tools - range of screwdrivers, pliers, shifters required as a lot of 
the existing tools are no longer on site

18-Sep 1 Yuasa RE12/7 battery for BBSS UPS
18-Sep 1 Liebert UPS and open comms card to replace multiple UPS units 

in the E Van.  If approved will also require several 110v power 
boards and a couple of extension cables to tidyup the wiring in the
van.

18-Sep Rag on a Roll (ROAR)
22-Sep Cable ties - various sizes


